Author's response to reviews

Title: A telephone survey of factors affecting willingness to participate in health research surveys

Authors:

Deborah C Glass (deborah.glass@monash.edu)
Helen L Kelsall (helen.kelsall@monash.edu)
Claudia Slegers (claudiamslegers@gmail.com)
Andrew B Forbes (andrew.forbes@monash.edu)
Bebe Loff (bebe.loff@monash.edu)
Deborah Zion (Deborah.Zion@vu.edu.au)
Lin Fritschi (lin.fritschi@curtin.edu.au)

Version: 8 Date: 24 September 2015

Author's response to reviews: see over
Editorial request:

a) please move the "List of Abbreviations", "Competing interest", "Author's contribution" and "Acknowledgement" before the "reference section" and after the "Conclusion section".
Done

b) Please move your 'funding' to the "Acknowledgement" section.
Done

c) Please move the "Ethics approval" in the "Methods" section.
Done